Author's response to reviews

Title: Non-neuronal and neuronal BACE1 elevation in association with angiopathic and leptomeningeal beta-amyloid deposition in the human brain

Authors:

Zhi-Qin Xue (kxglsjs@126.com)
Zheng-Wen He (hezhw2001@163.com)
Jian-Jun Yu (dryuijianjun@sina.com)
Yan Cai (caojicyan@csu.edu.cn)
Wen-Ying Qiu (gijuwy73@126.com)
Aihua Pan (panaihua@csu.edu.cn)
Wei-Ping Gai (weiping.gai@flinders.edu.au)
Huaibin Cai (caih@mail.nih.gov)
Xue-Gang Luo (luoxuegang@csu.edu.cn)
Chao Ma (mcpumc@163.com)
Xiao-Xin Yan (yanxiaoxin@csu.edu.cn)

Version: 5 Date: 17 April 2015

Author's response to reviews: see over
Dear Dr. Tim Shipley and Nhell:

Thank you very much for the further advice. We have removed the age and sex information of individual donors from Table 1, and indicated the total numbers of males and females, as well as age range in the two the groups in Column 1 of the table. The mean ages and standard derivations of the two groups are kept in the original text.

Thank you again for your consideration for publication of our work,

Xiao-Xin Yan

---

Instruction from editors:
MS: 6089250201549971
Non-neuronal and neuronal BACE1 elevation in association with angiopathic and leptomeningeal beta-amyloid deposition in the human brain
Zhi-Qin Xue, Zheng-Wen He, Jian-Jun Yu, Yan Cai, Wen-Ying Qiu, Aihua Pan, Wei-Ping Gai, Huaibin Cai, Xue-Gang Luo, Chao Ma and Xiao-Xin Yan

Dear Prof. Yan,

After assessing your manuscript, we note that table 1 still contains individual clinical data for patients included in your study. If patients provided written informed consent for the publication of individual data, then this must be stated in your manuscript. Otherwise, we would advise you to remove all individual age and sex information from table 1 to further anonymise patients.

............
With best wishes,

Nhell

Jhonell De Los Santos
on behalf of Dr. Tim Shipley
Journal Editorial Office
BioMed Central

e-mail: http://editorial@biomedcentral.com/
Web: http://www.biomedcentral.com/